Author's response to reviews

Title: Prediction of Intracranial Findings on CT-scans by Alternative Modelling Techniques

Authors:

Tjeerd van der Ploeg (tvploeg@quicknet.nl)
Marion Smits (marion.smits@erasusmc.nl)
Diederik W Dippel (d.dippel@erasusmc.nl)
Myriam Hunink (m.hunink@erasusmc.nl)
Ewout W Steyerberg (e.steyerberg@erasusmc.nl)

Version: 6 Date: 18 October 2011

Author's response to reviews: see over
We would like to thank you for your additional comments on our paper, MS: 3156340865403205 - Prediction of Intracranial Findings on CT-scans by Alternative Modelling Techniques, and for giving us the opportunity to revise it.

As you requested, we made the following major additional revisions to our manuscript.

**Major revisions**

**References**

We corrected the sequence of the references and inserted citations which were missing. We noticed that three references had been included by mistake and we eliminated these from the reference list. All 10 remaining references are now cited in numerical order.

**Figures**

Figure 4 (a, b and c) was changed into three separate figures numbered figure 4, 5 and 6. The numbering of the other figures was changed accordingly and all the references to the figures were changed in the manuscript.

**Figures**

We listed all the figures instead of using ranges.

We hope that these adaptations are in line with your expectations and look forward to your response to the re-submission of our manuscript.

On behalf of all the authors,

Yours sincerely

Prof. Dr. E.W. Steyerberg
Department of Public Health
Erasmus Medical Centre, Rotterdam